Author’s response to reviews

Title: Effectiveness of internet-delivered education and home exercise supported by behaviour change SMS on pain and function for people with knee osteoarthritis: A randomised controlled trial protocol.

Authors:

Rachel Nelligan (rachel.nelligan@unimelb.edu.au)
Rana Hinman (ranash@unimelb.edu.au)
Jessica Kasza (jessica.kasza@monash.edu)
Kim Bennell (k.bennell@unimelb.edu.au)

Version: 2 Date: 05 Jul 2019

Author’s response to reviews:

Friday 5th July 2019

Editor

BMC Musculoskeletal Disorders

Dear Editor,

Please find attached our updated application with

1. Figures
Uploaded

2. Clean manuscript
Provided

Sincerely,

Ms Rachel Nelligan
Centre for Health, Exercise and Sports Medicine
University of Melbourne, Australia